Author’s response to reviews

Title: Experiences of interactive ultrasound examination among women at risk of preterm birth: a qualitative study.

Authors:
Henrika Pulliainen (hpherm@utu.fi)
Hannakaisa Niela-Vilén (hmniel@utu.fi)
Eeva Ekholm (eeva.ekholm@tyks.fi)
Sari Ahlqvist-Björkroth (sarahl@utu.fi)

Version: 2 Date: 03 Sep 2019

Author’s response to reviews:
Lauren MacMillan
Editor
BMC Pregnancy and Childbirth

Thank you for the encouraging and positive comments about our manuscript.

We wish that you could find the revised paper suitable and interesting for publication Hereby we again state, that this is original work, has not been published previously and is not under consideration for publication elsewhere. If accepted for publication, it will not be published elsewhere including electronically in the same form, in English or in any other language, without the written consent of the copyright-holder. All authors also have approved the final, revised version of the manuscript. All authors declare that they have no conflicting interest to disclose.

This letter provides a detailed response to each point raised describing the amends done to the manuscript text and where they can be found.
1. We note that the order of authors has changed since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

In such cases, we use a standardised form which we would request you and your co-authors to complete. The authorship change form can be found at the following link: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5

All instructions can be found on the form, please treat the 'current authorship' section as the original authorship. Please return the form to the Editorial Office with all author signatures (including those newly added/removed), or attach it to your revised submission.

The order of authors was changed because we re-evaluated the authors' contributions. Eeva Ekholm and Sari Ahlqvist-Björkroth are both supervisors to PhD student Henrika Pulliainen and made equal contribution in the writing process of the manuscript. However, we evaluated that S.A-B. took slightly more responsibility in the early state of the conception and design of the study and in addition conducted all the interviews. Therefore, we all agree that the order of authors should be changed so that S.A-B. is the last author. However, we wish that Eeva Ekholm could share the last authorship with S.A-B.

Due to journey of one of the authors obtaining signatures turned out challenging by August 30th. The exact date of submitting revised manuscript was postponed to September 2nd.

2. The quotations used in your manuscript contain potentially identifying information. Please amend your manuscript by either limiting the amount of potentially-identifying information presented, or by obtaining explicit consent to publish such information.

To limit the identifying information, please remove the ages from your quotes to help maintain participant confidentiality. Please use age ranges in place of the ages. Please check that the identifiers do not link to patients.

In order to limit the identifying information, the numbered IDs and ages were removed from the quotes. We decided not replace ages with age ranges, because same information can be found in Table 3., Background characteristics.

3. In your 'Availability of Data and Materials' section, please change the word 'secrecy' to 'confidentiality'.

Word ‘secrecy’ was changed to ‘confidentiality’.
4. In line 81 of your manuscript, please change 'A convenient sample of pregnant women...' to 'A convenience sample of pregnant women...'.

Line 81 was changed as recommended.

5. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

Abbreviations were defined in the text at first use and a list of abbreviations can be found in the Declarations.

6. Please ensure that all supplementary files are cited within the main text (this does not include the declarations). Any items which are not cited may be deleted by our production department upon publication.

All the supplementary files are cited within the main text.

7. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

All the three tables can be found in the clean version and no additional files were loaded.

Thank you for your consideration and time.

Yours sincerely,

Henrika Pulliainen

MD, Doctoral researcher
tel. +358445274024
E-mail: hpherm@utu.fi
Address: Pampinkuja 4 B 7
20900 Turku